UC San Diego

UC San Diego Previously Published Works

Title

Real-world treatment utilization and economic implications of lupus nephritis disease activity in the United States.

Permalink https://escholarship.org/uc/item/6b05h75g

Journal Journal of Managed Care and Specialty Pharmacy, 29(1)

Authors

DallEra, Maria Eaddy, Michael Ogbonnaya, Augustina <u>et al.</u>

Publication Date

2023

DOI

10.18553/jmcp.2022.21496

Peer reviewed

Real-world treatment utilization and economic implications of lupus nephritis disease activity in the United States

Maria Dall'Era, MD; Kenneth Kalunian, MD; Michael Eaddy, PharmD, PhD; Augustina Ogbonnaya, MPH; Eileen Farrelly, MPH; Eric Turowski, MBA; Vanessa Birardi, PharmD; Neil Solomons, MD; Simrat Randhawa, MD; Paola Mina-Osorio, MD, PhD

Plain language summary

A common problem with lupus is damage in the kidneys (lupus nephritis). This damage can lead to the kidneys not working correctly (kidney failure). There are times of active disease in the kidneys and times when the disease is quiet (low disease activity). This study reviews the costs and drugs used in active disease, low disease activity, or kidney failure. The study found that drug costs are the highest during active disease or kidney failure.

Implications for managed care pharmacy

This is the first study to describe the economic implications of high and low disease activity and end-stage renal disease (ESRD) in lupus nephritis (LN). Monthly medical costs were doubled for high disease activity periods and were 7 times higher for ESRD vs low disease activity periods. Because active LN is evident in approximately two-thirds of patients, treatments that can achieve low disease activity remain a significant unmet need in the treatment of LN.

Author affiliations

Division of Rheumatology, Department of Medicine, School of Medicine, University of California, San Francisco (Dall'Era); Division of Rheumatology, Allergy, and Immunology, School of Medicine, University of California, San Diego, La Jolla (Kalunian); Xcenda, LLC, Carrollton, TX (Eaddy, Ogbonnaya, Farrelly), now with Abbvie, North Chicago, IL (Eaddy); Aurinia Pharmaceuticals, Inc., Rockville, MD (Turowski, Birardi, Solomons, Randhawa, Mina-Osorio).

AUTHOR CORRESPONDENCE: Paola Mina-Osorio, 917.561.2753; Paola.Mina-osorio@immunovant.com

> J Manag Care Spec Pharm. 2023;29(1):36-45 Copyright©2023, Academy of Managed Care Pharmacy. All rights reserved.

ABSTRACT

BACKGROUND: Lupus nephritis (LN) is a common and severe complication of systemic lupus erythematosus (SLE), with approximately 40% of patients with SLE developing LN. Even with treatment, 10%-30% of patients will progress to end-stage renal disease (ESRD). Although many studies have assessed the clinical value of low disease activity in LN, the economic implications are less defined.

OBJECTIVE: To evaluate treatment utilization and health care costs associated with active

disease, low disease activity, and ESRD in patients with LN.

METHODS: A retrospective analysis of Optum pharmacy and medical claims data from 2015 to 2019 was performed and included patients with a diagnosis of SLE (International Classification of Diseases, Ninth Revision or Tenth Revision codes 710.0 or M32, respectively) and additional prespecified criteria for LN. Total health care payer costs for medical and pharmacy services and treatment utilization for commonly prescribed medications were determined for periods of low disease activity, active disease, or ESRD. **RESULTS:** A total of 21,251 patients (mean age 60.3 years; 87% female; 55% White patients and 18% Black patients) with a mean follow-up period of 30.6 months were included; the majority of patients had active disease (67.3%), followed by low disease activity (51.3%), and ESRD (10.5%). Glucocorticoids were used 2 times more often and mycophenolate mofetil was used 4 times more often in patients with active disease vs low disease activity. Glucocorticoids, mycophenolate mofetil, and tacrolimus were more commonly used in patients with ESRD vs those with low disease activity. Mean medical costs were \$4,777 per month in active disease and \$18,084 per month in ESRD vs \$2,523 per month in low disease activity.

CONCLUSIONS: Treatment burden and costs are high for patients with active disease and ESRD in LN. Treatments that allow patients to achieve and maintain low disease activity may help improve patient outcomes and reduce medication use and overall health care costs.

Lupus nephritis (LN) is a common and severe complication of systemic lupus erythematosus (SLE) that can cause irreversible nephron loss, thus reducing the life span of the kidneys, and may ultimately lead to end-stage renal disease (ESRD).¹ Patients diagnosed with LN have 3 times the risk of mortality compared with patients with SLE and no renal involvement.² Approximately 40% of patients with SLE will develop LN,3 and 35% will present with LN at initial diagnosis of SLE.⁴ LN features alternating periods of active disease (ie, flares or relapses) and low disease activity. Potent initial therapy with moderate- to high-dose glucocorticoids and mycophenolate mofetil (MMF) or cyclophosphamide is used during active disease with the goal of suppressing the immune-mediated inflammation, whereas subsequent therapy using lower-dose MMF or azathioprine is aimed at consolidating the initial response, maintaining the renal response, and preventing flares.5

However, even with treatment, 10%–30% of patients with LN will progress to ESRD within 15 years of diagnosis and will require dialysis or kidney transplant.⁶ For this reason, one of the main goals of treatment is to preserve long-term kidney function by achieving complete response,^{7,8} which includes a reduction in proteinuria (urine protein creatinine ratio <0.5 to 0.7 gm per 24 hours) and improvement or stabilization of estimated glomerular filtration rate.⁷ Decreases in proteinuria within 6 to 12 months of initiating treatment are associated with better long-term outcomes, such as decreased risk of mortality and ESRD.⁹⁻¹⁵

Although the clinical value of achieving low disease activity has been well documented,⁹⁻¹⁵ the economic implications are less well defined, and only a few studies have assessed real-world treatment of LN. Previous studies had small sample sizes (range: 900-2,300 patients)¹⁶⁻¹⁹ or analyzed outcomes in specific populations, such as patients enrolled in Medicaid.¹⁸ A recent prospective study in patients with SLE determined that spending more than 50% of time in low disease activity was associated with a 47% reduction in annual direct medical costs.²⁰ Although that study evaluated the impact of low disease activity on costs, no such studies have been conducted in patients with LN specifically.

Thus, the purpose of this study was to evaluate treatment utilization and health care payer costs associated with active disease, low disease activity, and ESRD in patients with LN.

Methods

STUDY DESIGN AND DATA SOURCE

This descriptive study was a retrospective, observational analysis using Optum pharmacy and medical claims data dated between January 1, 2007, and December 31, 2019 (Figure 1). The Optum database includes US health care plan data on more than 111 million patients with commercial or Medicare with Part D coverage and includes demographic information, International Classification of Diseases, Ninth Revision and Tenth Revision (ICD-9; ICD-10) codes, Current Procedural Terminology codes, Healthcare Common Procedure Coding System codes for inpatient and outpatient diagnoses and procedures, and prescription information such as National Drug Code. Primary data were not collected in this study; all data were ascertained from the previously anonymized Optum pharmacy and medical claims data. Therefore, institutional review board approval and informed consent were not required. Patients enrolled in Medicaid were not captured in this database.

STUDY POPULATION

Patients with evidence of any diagnosis of SLE (ICD-9: 710.0 or ICD-10: M32) with 1 or more inpatient admission or 2 or more outpatient admissions 30 days apart between January 1, 2015, and December 31, 2019, were included in the study. Because of the nonspecificity of ICD-9 codes in identifying LN (ICD-9 codes were used prior to October 1, 2015), and because LN is often undercoded in claims data, inclusion criteria from previous studies that evaluated LN in administrative claims data were evaluated by our clinical experts and adopted for patient identification.^{16,21} Patients had to meet 1 or more of the following criteria for inclusion in the analysis: 2 or more outpatient or 1 or more inpatient events associated with a diagnosis of glomerular disease in SLE (ICD-10: M32.14), a diagnosis of SLE with at least 1 nephrology visit, a diagnosis of an acute or chronic renal condition, evidence of ESRD (eg, diagnosis code, dialysis, use of phosphate binder or renal transplant) after SLE diagnosis, or use of medication for treatment of LN (see Supplementary Table 1, available in online article, for full criteria and codes).

Patients were categorized as prevalent or incident, in which incident cases were defined as having 6 months of eligibility without evidence of LN before the first LN



diagnosis, which occurred after January 1, 2015 (Figure 1). Prevalent cases were defined as having evidence of LN prior to January 1, 2015. The index date for prevalent cases was the first month of eligibility during the study period, whereas the index date for incident cases was the date the patient met the LN case definition during the study period. The follow-up (post-index) period included the period following the index date, which was used to assess costs and treatment utilization. Adult patients (\geq 18 years) with at least 2 months of continuous enrollment were followed until death, loss to follow-up, or the end of the study period.

FOLLOW-UP CLASSIFICATIONS AND ENDPOINT DEFINITIONS

Patient follow-up time was classified as periods of low disease activity, active disease, or ESRD; thus, patients were not mutually exclusive to each cohort. Low disease activity was evaluated across each patient's follow-up period and was defined as at least 6 consecutive months of glucocorticoid dose less than or equal to 5 mg per day, MMF dose less than or equal to 2 gm per day, and no use of cyclophosphamide. For the glucocorticoid dose, the prednisone equivalent dose was calculated and included oral and intravenous administration. Time without evidence of low disease activity or ESRD was classified as active disease. Patients were defined as being in ESRD if there was evidence of at least 1 ESRD diagnosis code, use of dialysis, treatment with phosphate binders, or kidney transplant.

The primary endpoints included total health care payer costs for all-cause medical and pharmacy services during periods of active disease, low disease activity, and ESRD and treatment utilization of commonly prescribed medications for LN. Treatment utilization was based on claims for prescription drugs dispensed (retail and mail order) and included all glucocorticoids (intravenous and oral), MMF, azathioprine, cyclophosphamide, calcineurin inhibitors (cyclosporine and tacrolimus), rituximab, and belimumab. Frequency and time to mortality were secondary endpoints.

Health care payer costs were defined as the total amount paid for all-cause medical and pharmacy services, including costs related to LN and conditions other than LN. Costs were categorized as pharmacy, inpatient visits, emergency department visits, physician office visits, and other outpatient services (eg, laboratory services, outpatient treatment, diagnostic tests). LN-related medications billed through the medical benefit were classified as pharmacy costs and included all medications included in the prescription claims analysis. Given the differential follow-up time across patients, costs were presented as monthly costs per patient and adjusted to 2020 US dollars.

Additionally, the frequency of use of long-term glucocorticoids (LTGs) was evaluated in a subset of patients who were followed for at least 12 consecutive months. LTG use was defined as evidence of glucocorticoid use for at least 6 months, with a medication possession ratio of 0.80 or more to account for potential nonadherence (1.0 represents complete adherence). Medication possession ratio was defined as the number of days of therapy during evidence of LTG divided by the maximum number of possible days between the first and last qualifying glucocorticoid prescription. The assessment of LTG was not restricted to specific disease activity time frames, as patients could have evidence of LTG in multiple disease states (ie, active disease, low disease activity, or ESRD).

Additionally, the Quan-Charlson Comorbidity Index (CCI) and data on selected comorbidities, including diabetes (with and without end-organ damage), myocardial infarction, congestive heart failure, chronic pulmonary disease, cerebrovascular disease, all tumors (including leukemia and lymphoma), infections (influenza, acute bronchitis, pneumonia, bacterial, or urinary tract infection), hypertension, gastrointestinal symptoms (bleeding; diarrhea; cramping; or bloody, black, or tarry stool), psychiatric issues (cognitive impairment, memory loss, dementia, or insomnia), skin conditions (ie, rashes, purpura), edema, and osteoporosis were collected, as they are associated with LTG use.

STATISTICAL METHODOLOGY

Baseline and demographic patient characteristics were analyzed descriptively using percentages, means, and SDs in patients by active disease, low disease activity, and ESRD, including all patients who met the criteria of the specified period at least once during the study. Comorbidities were assessed using the Quan-CCI score. Endpoint results are presented descriptively for periods of active disease, low disease activity, and ESRD. P values were not calculated for comparisons. Differences between patients with commercial and Medicare with Part D coverage are highlighted.

Results

There were 101,972 patients with evidence of SLE from January 2007 to October 2019. Of these patients, 38,679 (37.9%) had evidence of LN, with 28,891 who were present in the database as of or after January 1, 2015 (see <u>Supplementary Figure 1</u>). After applying eligibility criteria for the study, the final sample size was 21,251, retaining 73.6% of the eligible sample (see <u>Supplementary Table 1</u> for the classification of patients meeting each criterion for a diagnosis of LN). A total of 53.6% (n=11,380) of the eligible sample was defined as incident cases and 46.5% (n=9,871) was defined as prevalent cases. Mean follow-up time of the sample was 30.6 months.

BASELINE PATIENT CHARACTERISTICS

The majority of patients were female (86.9%), White (55.2%), and from the South (47.1%) (Table 1). Low disease activity was evident in 51.3% of patients, with a mean duration of 27.5 months. There was evidence of active disease in 67.3% of patients, with a mean duration of 20.5 months. ESRD was evident in 10.5% of patients; mean time to ESRD was 18 months from study index.

Patients having evidence of low disease activity were older (mean age: 62.4 years) when compared with patients with active disease (mean age: 59.4 years) or ESRD (mean age: 58.0 years). The older age across all cohorts was driven by a high Medicare population, with 61.9% of the sample being enrolled in Medicare (mean age: 67.9 years) compared with 38.1% enrolled in commercial insurance (mean age: 47.8 years). Although the distributions of sex and race were similar in patients with low disease activity and active disease, ESRD had a lower proportion of females (ESRD: 80.6%; active disease: 87.7%; low disease activity: 87.7%) and White patients (ESRD: 49.6%; active disease: 55.2%; low disease activity: 57.1%).

TABLE 1 Baseline Patient Characteristics								
Patient characteristic	Overall sample N=21,251		Low disease activity n=10,911		Active disease n=14,310		End-stage renal disease n=2,240	
Age, mean±SD	60.3	±15.7	62.4±15.1		59.4±15.7		58.0±15.7	
Quan-CCI, mean±SD	2.35±2.01		2.20±1.90		2.30±2.00		2.90±2.30	
Sex								
Female	18,470	(86.9)	9,570	(87.7)	12,555	(87.7)	1,805	(80.6)
Male	2,774	(13.1)	1,337	(12.3)	1,752	(12.2)	434	(19.4)
Unknown	7	(0)	4	(0)	3	(0)	1	(0)
Race and ethnicity								
White	11,733	(55.2)	6,234	(57.1)	7,893	(55.2)	1,110	(49.6)
Black	3,734	(17.6)	1,808	(16.6)	2,550	(17.8)	550	(24.6)
Hispanic	3,089	(14.5)	1,604	(14.7)	2,078	(14.5)	333	(14.9)
Asian	666	(3.1)	324	(3.0)	445	(3.1)	82	(3.7)
Unknown	2,029	(9.6)	941	(8.6)	1,344	(9.4)	165	(7.4)
Comorbidities								
Hypertension	12,928	(60.8)	6,597	(60.5)	8,567	(59.9)	1,521	(67.9)
Gastrointestinal symptoms	7,024	(33.1)	2,471	(22.7)	3,708	(25.9)	589	(26.3)
Any infection	6,491	(30.5)	3,022	(27.7)	4,506	(31.5)	738	(33.0)
Diabetesª	6,231	(29.3)	3,233	(29.6)	4,029	(28.2)	804	(35.9)
Chronic pulmonary disease	2,614	(12.3)	1,408	(12.9)	1,966	(13.7)	344	(15.4)
Psychiatric events	2,611	(12.3)	1,162	(10.7)	1,566	(10.9)	191	(8.5)
Edema	2,609	(12.3)	1,178	(10.8)	1,787	(12.5)	361	(16.1)
Congestive heart failure	2,547	(12.0)	1,165	(10.7)	1,660	(11.6)	409	(18.3)
Osteoporosis	2,470	(11.6)	1,184	(10.9)	1,702	(11.9)	211	(9.4)
Cerebrovascular disease	1,809	(8.5)	910	(8.3)	1,133	(7.9)	233	(10.4)
Skin conditions	1,526	(7.2)	685	(6.3)	1,054	(7.4)	112	(5.0)
Any tumors ^ь	1,413	(6.7)	665	(6.1)	987	(6.9)	151	(6.7)
Myocardial infarction	879	(4.1)	395	(3.6)	566	(4.0)	108	(4.8)
Region								
Northeast	1,938	(9.1)	990	(9.1)	1,255	(8.8)	193	(8.6)
Midwest	3,719	(17.5)	1,814	(16.6)	2,529	(17.7)	405	(18.1)
South	10,016	(47.1)	4,885	(44.8)	6,925	(48.4)	1,203	(53.7)
West	5,542	(26.1)	3,205	(29.4)	3,576	(25.0)	433	(19.3)
Other/unknown	36	(0.2)	17	(0.2)	25	(0.2)	6	(0.3)
Payer type								
Commercial	8,101	(38.1)	3,645	(33.4)	5,642	(39.4)	930	(41.5)
Medicare	13,150	(61.9)	7,266	(66.6)	8,668	(60.6)	1,310	(58.5)

Values are the number (percentage), unless otherwise indicated. Baseline characteristics were assessed during the 6-month pre-index period, including the index date.

^aWith and without organ damage.

^bIncluding leukemia and lymphoma.

Quan-CCI=Quan-Charlson Comordity Index.

TABLE 2 Treatment Utilization for Active Disease, Low Disease, and End-Stage Renal Disease								
Treatment	Overal N=2	l sample 1,251	Low disea n=1	ase activity .0,911	vity Active disease n=14,310		End-stage renal disease n=2,240	
Glucocorticoid	16,043	(75.5)	4,741	(43.5)	12,789	(89.4)	1,563	(69.8)
Antihypertensive	11,621	(54.7)	5,721	(52.4)	7,444	(52.0)	1,180	(52.7)
Statin	8,696	(40.9)	4,462	(40.9)	5,393	(37.7)	934	(41.7)
Mycophenolate mofetil	2,538	(11.9)	415	(3.8)	2,007	(14.0)	430	(19.2)
Azathioprine	1,697	(8.0)	521	(4.8)	1,289	(9.0)	148	(6.6)
Rituximab	979	(4.6)	214	(2.0)	790	(5.5)	66	(3.0)
Belimumab	675	(3.2)	180	(1.7)	570	(4.0)	33	(1.5)
Tacrolimus	478	(2.3)	37	(0.3)	143	(1.0)	333	(14.9)
Cyclophosphamide ^a	427	(2.0)		-	337	(2.4)	67	(3.0)
Cyclosporine	165	(0.8)	33	(0.3)	83	(0.6)	61	(2.7)
Values are the number (percen	itage), unless oth	erwise indicated			·	·	÷	·

"No use of cyclophosphamide was a criterion for low disease activity.

Baseline comorbidities were assessed in the 6-month pre-index period, including the index date, by the Quan-CCI and by evaluating specific comorbidities associated with LN. The CCI was similar in patients with active disease (2.3) and low disease (2.2), whereas ESRD patients had the highest CCI (2.9); 70.2% of ESRD patients had a CCI greater than or equal to 2 compared with roughly 55% of patients in the active and low disease cohorts. When evaluating the distribution of comorbidities associated with LN, patients having low disease activity did not have the highest risk in any comorbidity category. Patients with low disease activity tended to have a similar or slightly lower risk of comorbidities when compared with active disease patients, whereas patients with ESRD had the highest risk of hypertension, diabetes, edema, congestive heart failure, and cerebrovascular disease.

TREATMENT UTILIZATION

As seen in Table 2, although glucocorticoids were the most frequently used medication across all cohorts, use of glucocorticoids was observed in almost twice as many patients with active disease (89.4%) than those with low disease activity (43.5%). With the exceptions of antihypertensives and statins, low disease patients tended to have the lowest rate of LN treatments. Antihypertensive (52.0%-52.7%) and statin (37.7%-41.7%) use was similar across the cohorts. The rate of MMF use was approximately 4 times higher in active disease patients (14.0%) and 5 times higher in ESRD patients (19.2%) compared with low disease activity patients (3.8%). When comparing MMF rates by insurance types, Medicare patients were less likely to have evidence of MMF use (6.3%) compared with commercially insured patients (21.0%). Tacrolimus was used in 14.9% of ESRD patients and less than or equal to 1.0% of patients with active disease or low disease activity.

LTG use (≥ 6 months) was not assessed within specific disease time frames (ie, the cohorts were not mutually exclusive). Accordingly, 17.2% of patients with active disease had evidence of LTG use compared with 10.7% of patients with low disease activity and 5.5% of ESRD patients.

HEALTH CARE COSTS

All-cause medical costs were almost twice as high during active disease periods, with a mean of \$4,777 per month compared with \$2,523 per month for periods of low disease activity (Table 3). Costs associated with inpatient, emergency department, physician office, and other outpatient visits were lower for patients with low disease activity vs active disease, with the greatest difference seen between inpatient costs (\$1,406 vs \$2,896). Similarly, pharmacy costs were lower for periods of low disease activity (\$1,061) compared with high disease activity (\$1,835). After factoring in pharmacy costs, monthly total health care payer costs for active disease periods were \$3,028 higher per month compared with periods of low disease activity.

The mean monthly all-cause medical cost for patients with ESRD was \$18,084 (Table 3). The main medical cost drivers were inpatient visits (\$13,756) and other outpatient visits (\$3,013). Monthly pharmacy costs were \$3,760, bringing mean monthly total payer costs for all health care to \$21,844 for patients with ESRD.

TABLE 3

Health Care Costs for Active Disease, Low Disease Activity, and End-Stage Renal Disease

Monthly cost, mean	Low disease activity n=10,911	Active disease n=14,310	End-stage renal disease n=2,240
Medical	\$2,523	\$4,777	\$18,084
Inpatient	\$1,406	\$2,896	\$13,756
Emergency department	\$281	\$657	\$663
Physician office	\$304	\$457	\$655
Other outpatient	\$532	\$768	\$3,013
Pharmacy	\$1,061	\$1,835	\$3,760
Total	\$3,584	\$6,612	\$21,844
Medicare	n = 7,266	n=8,668	n=1,310
Medical	\$2,889	\$5,124	\$20,457
Pharmacy	\$1,095	\$1,781	\$4,793
Total	\$3,984	\$6,905	\$25,250
Commercial	n=3,645	n=5,642	n=930
Medical	\$1,794	\$4,243	\$14,741
Pharmacy	\$993	\$1,920	\$2,305
Total	\$2,787	\$6,163	\$17,047

When costs were stratified by type of insurance, the overall mean monthly all-cause health care cost was slightly higher for Medicare patients (\$6,305) compared with commercial patients (\$5,057). Regardless of insurance type, lower costs were seen in low disease activity compared with active disease and ESRD (Table 3). The difference in total monthly costs with active vs low disease activity for Medicare patients was \$2,921 and for commercial patients was \$3,376. Total monthly all-cause health care costs for Medicare patients in ESRD were \$25,250; the total costs for commercial patients were \$17,047.

DEATH

Overall, 11.6% of patients (n=2,462) died over the followup period. Mean time to death was 23.7 months after study index. The rate of death in patients with ESRD was 24.3% compared with 11.0% in active and 10.0% in low disease patients.

Discussion

This is the first and largest study, to our knowledge, that captures real-world treatment utilization and economic implications associated with low disease activity, active disease, and ESRD in patients with LN in a US population. Previous studies in LN have described treatment utilization and associated costs of care but did not specifically assess differences between disease activity levels. Results from the current study indicated that patients with active disease have higher health care costs than patients with low disease activity (\$6,612 vs \$3,584) and supports that these patients have high treatment burden, including high glucocorticoid use, as they are initiating therapy for LN or treating a flare. Given that 67.3% of patients had evidence of active disease, treatments that help patients achieve and maintain low disease activity remain a significant unmet need within the treatment landscape for LN.

Relative to low disease activity, the economic burden for patients with active disease and ESRD was substantial. Monthly all-cause medical costs for periods of active disease were twice as high and costs for ESRD were 7 times as high when compared with periods of low disease activity. The death rate associated with ESRD (24%) was also more than twice as high compared with low disease activity (10%) and active disease (11%). As previous studies have correlated disease status in LN with long-term renal survival, our results indicate that early intervention and rapid management of active disease periods to help achieve lower disease activity would reduce the economic burden and reduce the risk of mortality associated with the development of ESRD.

In the current study, 51.3% of patients had evidence of low disease activity during follow-up. Although low disease activity was defined in this study by at least 6 consecutive months of glucocorticoid dose less than or equal to 5 mg per day, MMF dose less than or equal to 2 gm per day, and no use of cyclophosphamide, this percentage aligns with previous research, which reported that 40.9%-59.1% of patients had low disease activity at 24 months post-LN diagnosis, as assessed by clinical response based on various laboratory measures of renal function and activity.¹¹ Additionally, 10% of patients in our study had evidence of ESRD, which aligns with previous estimated ranges of 10%-30%.⁶

This study also provided updated information on treatment utilization of medications used in LN and showed a higher treatment burden for patients with active disease and ESRD. Regardless of disease status, glucocorticoids were the most frequently used medication, being more heavily used in periods of active disease (89.4%) and ESRD (69.8%) compared with low disease activity (43.5%). The frequent use of glucocorticoids in active disease is consistent with guidelines for treating kidney flares in LN⁷; glucocorticoids are also used as maintenance therapy in patients who are candidates or have received kidney transplantation.²² Long-term use of glucocorticoids is associated with many adverse effects, including some that can be quite costly, such as serious infections, osteoporotic fractures, diabetes mellitus, psychiatric events, and premature mortality.^{3,23} As such, there has been a recent focus to reduce glucocorticoid dose and rapidly taper after 6 months in patients with LN.⁷ Evidence of LTG use was highest in patients with active disease.

MMF use was approximately 4 times higher in active disease patients and 5 times higher in ESRD patients compared with low disease activity, and tacrolimus use was substantially higher (~15%) in ESRD patients compared with active disease or low disease activity patients ($\leq 1\%$); this may be related to the use of tacrolimus as maintenance therapy for kidney transplantation.22 The overall rate of MMF use was lower than expected (11.9%) in our study. In the study by Bartels-Peculis et al, 30% of patients had evidence of MMF use; however, this population was substantially younger, with a mean age of 47 years, compared with patients in the current study, who had a mean age of 60 years.¹⁶ An earlier study using data from a national managed care organization with 907 LN patients showed similar results to Bartels-Peculis et al, with 50% of the overall cohort aged between 18 and 44 years and 21% using MMF.¹⁷ In a comparison of MMF rates by insurance type in the current study, commercially insured patients were twice as likely to have evidence of MMF use (21.0%) compared with patients with Medicare with Part D coverage (9.4%). It is possible that patients may have received medications such as MMF through prescription assistance programs or coupon discount programs, such as GoodRx or NeedyMeds, Inc., that would not be captured in the dataset used for this study.

The rates of use of antihypertensives and statins were similar across cohorts, with about 50% and 40% of all patients using each, respectively. The most recent American College of Rheumatology guidelines for LN recommend antihypertensives for all LN patients with proteinuria greater than or equal to 0.5 gm per 24 hours (or equivalent by protein/creatinine ratios on spot urine samples) and statin therapy for patients with low-density lipoprotein cholesterol >100 mg/dL.⁵ These rates of antihypertensive and statin use are expected given the comorbidity rates in this study sample for hypertension (60.8%) and cardiovascular disease (12.3% for chronic pulmonary disease, 8.5% for congestive heart failure, 4.1% for myocardial infarction).

One advantage of the current study is that it used a much shorter minimum follow-up period than previous studies on treatment utilization and costs.¹⁶⁻¹⁹ The current study required at least 2 months of follow-up after the index date; however, most previous studies have imposed a 12-month follow-up minimum following LN index date, which substantially reduces sample size and potentially

the generalizability of results.¹⁶⁻¹⁹ Patients with LN may die within the year after being identified for study inclusion or could transition to Medicare fee-for-service; therefore, exclusion of these patients may significantly underestimate the true costs incurred by payers. For example, the recent Bartels-Peculis study that imposed such restrictions indicated that total mean all-cause costs were \$3,789 per month (\$45,469 per year),¹⁶ whereas the current study showed total mean all-cause costs of \$5,829 per month. Mean follow-up for the current study was approximately 2.5 years; therefore, allowing a 2-month minimum follow-up did not shorten follow-up. The use of a variable follow-up time maximizes the opportunity to capture periods of active disease and low disease activity and correspondingly assesses how patients are treated during these periods. The inclusion criteria for this analysis were broad and could include patients with at least 1 diagnosis of a renal condition based on ICD-9 codes (Supplementary Table 1) instead of at least 2, as seen in a previous study.²¹ A sensitivity analysis was performed excluding patients with only 1 diagnosis code, which showed that excluding those patients had minimal impact on the costs and the observed trends remained the same.

LIMITATIONS

Research conducted with claims data has inherent limitations, including possible coding errors, lack of information on treatments obtained outside of an insurance setting, lack of detailed clinical information (eg, disease severity) and treatment adherence, and reasons underlying choice and provision of treatment. As noted, it is possible that patients may have received medications through other means that may not have been captured in the claims data; therefore, the results of the current study may be an underestimation of medication use. As disease activity periods were defined by observed treatment utilization and not clinical or serological parameters, decreased use of steroids, MMF, and cyclophosphamide in the low disease arm is expected. Additionally, missing medication claims may have caused patients to be misclassified with low disease activity, which means the cost differences observed in this study may be a conservative estimate of the economic implications of low disease activity. However, disease state estimates, as defined by treatment utilization in the current study, do align with clinically defined disease states.11 Additionally, some treatments used for LN are also used for SLE and in kidney transplantation; thus, it is difficult to extricate which treatments were used specifically for LN. Our study included a large Medicare insurance population, and the design of the study required these patients to have Medicare Part D for prescription coverage. Results may not be generalizable to the overall Medicare population. Many patients with ESRD have Medicare Part B coverage with or without Part D; as such, the study may not be reflective of Medicare patients without Part D coverage.

The study population consisted of commercially insured and Medicareinsured patients in the Optum database, and results may not be generalizable to the overall LN population. Future studies should be conducted in a Medicaid population for comparison. Almost 40% of patients initially identified in the database with SLE had evidence of LN, consistent with earlier reports.24 Rigorous inclusion criteria further reduced the studied population to 21,251 patients. Though LN affects primarily Black and Hispanic patients,24 the current sample was predominantly White patients (55.2%), with only 17.6% Black and 14.5% Hispanic patients. The mean age of the sample was 60 years, higher than other studies. Additionally, treatment utilization may differ according to disease duration, provider type, patient preference, and other characteristics that were not measured in the current study. The current study only assessed costs incurred to payers and did not assess patient costs for medications and services or indirect costs of LN (eg, work loss and productivity). Future research should seek to quantify the full cost of illness related to LN.

Conclusions

In summary, treatment burden and costs associated with active disease and ESRD in LN are high in comparison with low disease activity. Glucocorticoid and MMF use are more frequent in patients with active disease and ESRD, tacrolimus use is much more common in ESRD patients, and LTG use is higher in patients with active disease. Additionally, the mortality rate is twice as high for ESRD patients. Mean monthly allcause medical costs were much lower for periods of low disease activity, as costs were twice as high for periods of active disease and 7 times as costly for ESRD. Given the high rates of patients with active disease and poor outcomes associated with active disease, there remains significant unmet need in the treatment landscape for LN. The current study suggests that achieving low disease activity in LN patients is associated with reduced economic burden to health care payers; as such, treatments that allow patients to achieve and maintain low disease activity may help improve patient outcomes and reduce overall health care costs.

DISCLOSURES

Maria Dall'Era and Kenneth Kalunian are consultants of Aurinia Pharmaceuticals. Eric Turowski, Vanessa Birardi, Neil Solomons, Simrat Randhawa, and Paola Mina-Osorio are employees and stockholders of Aurinia Pharmaceuticals. Michael Eaddy is a former employee of Xcenda, LLC. Augustina Ogbonnaya and Eileen Farrelly are employees of Xcenda, LLC, which was contracted by Aurinia Pharmaceuticals to assist in the conduct of this study and the writing of this manuscript.

Aurinia Pharmaceuticals provided funding for this study and the preparation of the manuscript. Aurinia Pharmaceuticals had a role in writing the report and decision to submit for publication.

ACKNOWLEDGMENTS

The authors thank Tim Hermes, formerly of Aurinia Pharmaceuticals, for his review of the manuscript. The authors additionally thank Robin Tan, PhD, and Andrea Gundlach, PharmD, MPH, CMPP, of Xcenda, LLC for providing medical writing support and Vishal Saundankar of Xcenda, LLC for analytic support.

REFERENCES

1. Anders HJ, Rovin B. A pathophysiologybased approach to the diagnosis and treatment of lupus nephritis. *Kidney Int.* 2016;90(3):493-501. <u>doi:10.1016/j.</u> <u>kint.2016.05.017</u>

2. Hanly JG, O'Keeffe AG, Su L, et al. The frequency and outcome of lupus nephritis: results from an international inception cohort study. *Rheumatology* (*Oxford*). 2016;55(2):252-62. <u>doi:10.1093/</u> <u>rheumatology/kev311</u>

3. Hoover PJ, Costenbader KH. Insights into the epidemiology and management of lupus nephritis from the US rheumatologist's perspective. *Kidney Int.* 2016;90(3):487-92. <u>doi:10.1016/j.</u> <u>kint.2016.03.042</u>

4. Izmirly PM, Wan I, Sahl S, et al. The incidence and prevalence of systemic lupus erythematosus in New York County (Manhattan), New York: The Manhattan Lupus Surveillance Program. Arthritis Rheumatol. 2017;69(10):2006-17. doi:10.1002/art.40192

5. Anders H-J, Lei Y, Rovin BH. Induction and maintenance therapy of lupus nephritis: an obituary. *Kidney Int*. 2021;99(2):288-91. <u>doi:10.1016/j</u>. <u>kint.2020.11.009</u>

6. Bertsias GK, Tektonidou M, Amoura Z, et al; European League Against Rheumatism and European Renal Association-European Dialysis and Transplant Association. Joint European League Against Rheumatism and European Renal Association-European Dialysis and Transplant Association (EULAR/ERA-EDTA) recommendations for the management of adult and paediatric lupus nephritis. Ann Rheum Dis. 2012;71(11):1771-82. <u>doi:10.1136/</u> annrheumdis-2012-201940

7. Fanouriakis A, Kostopoulou M, Cheema K, et al. 2019 Update of the Joint European League Against Rheumatism and European Renal Association-European Dialysis and Transplant Association (EULAR/ERA-EDTA) recommendations for the management of lupus nephritis. Ann Rheum Dis. 2020;79(6):713-23. doi:10.1136/annrheumdis-2020-216924 8. Parikh SV, Almaani S, Brodsky S, Rovin BH. Update on lupus nephritis: core curriculum 2020. *Am J Kidney* Dis. 2020;76(2):265-81. <u>doi:10.1053/j.</u> ajkd.2019.10.017

9. Chen YE, Korbet SM, Katz RS, Schwartz MM, Lewis EJ; Collaborative Study Group. Value of a complete or partial remission in severe lupus nephritis. *Clin J Am Soc Nephrol.* 2008;3(1):46-53. doi:10.2215/CJN.03280807

10. Dall'Era M, Cisternas MG, Smilek DE, et al. Predictors of long-term renal outcome in lupus nephritis trials: Lessons learned from the Euro-Lupus Nephritis cohort. Arthritis Rheumatol. 2015;67(5):1305-13. <u>doi:10.1002/art.39026</u>

11. Davidson JE, Fu Q, Ji B, et al. Renal remission status and long-term renal survival in patients with lupus nephritis: A retrospective cohort analysis. J *Rheumatol.* 2018;45(5):671-7. <u>doi:10.3899/</u> jrheum.161554

12. Mackay M, Dall'Era M, Fishbein J, et al. Establishing surrogate kidney end points for lupus nephritis clinical trials: Development and validation of a novel approach to predict future kidney outcomes. Arthritis Rheumatol. 2019;71(3):411-9. <u>doi:10.1002/art.40724</u>

13. Tamirou F, D'Cruz D, Sangle S, et al; MAINTAIN Nephritis Trial Group. Long-term follow-up of the MAINTAIN Nephritis Trial, comparing azathioprine and mycophenolate mofetil as maintenance therapy of lupus nephritis. Ann Rheum Dis. 2016;75(3):526-31. <u>doi:10.1136/</u> annrheumdis-2014-206897 14. Tamirou F, Lauwerys BR, Dall'Era M, et al; MAINTAIN Nephritis Trial Investigators. A proteinuria cut-off level of 0.7 g/day after 12 months of treatment best predicts long-term renal outcome in lupus nephritis: Data from the MAINTAIN Nephritis Trial. Lupus Sci Med. 2015;2(1):e000123. doi:10.1136/ lupus-2015-000123

15. Ugolini-Lopes MR, Seguro LPC, Castro MXF, et al. Early proteinuria response: A valid real-life situation predictor of long-term lupus renal outcome in an ethnically diverse group with severe biopsy-proven nephritis? *Lupus Sci Med.* 2017;4(1):e000213. <u>doi:10.1136/</u> <u>lupus-2017-000213</u>

16. Bartels-Peculis L, Sharma A, Edwards AM, Sanyal A, Connolly-Strong E, Nelson WW. Treatment patterns and health care costs of lupus nephritis in a United States payer population. Open Access Rheumatol. 2020;12:117-24. doi:10.2147/OARRR.S248750

17. Furst DE, Clarke A, Fernandes AW, et al. Medical costs and healthcare resource use in patients with lupus nephritis and neuropsychiatric lupus in an insured population. J Med Econ. 2013;16(4):500-9. doi:10.3111/13696998.2013.772058

18. Li T, Carls GS, Panopalis P, Wang S, Gibson TB, Goetzel RZ. Long-term medical costs and resource utilization in systemic lupus erythematosus and lupus nephritis: A five-year analysis of a large medicaid population. Arthritis Rheum. 2009;61(6):755-63. <u>doi:10.1002/art.24545</u> 19. Pelletier EM, Ogale S, Yu E, Brunetta P, Garg J. Economic outcomes in patients diagnosed with systemic lupus erythematosus with versus without nephritis: Results from an analysis of data from a US claims database. *Clin Ther.* 2009;31(11):2653-64. <u>doi:10.1016/j.</u> <u>clinthera.2009.11.032</u>

20. Yeo AL, Koelmeyer R, Kandane-Rathnayake R, et al. Lupus low disease activity state and reduced direct health care costs in patients with systemic lupus erythematosus. *Arthritis Care Res* (Hoboken). 2020;72(9):1289–95. doi:10.1002/ acr.24023

21. Chibnik LB, Massarotti EM, Costenbader KH. Identification and validation of lupus nephritis cases using administrative data. *Lupus*. 2010;19(6): 741-3. <u>doi:10.1177/0961203309356289</u>

22. Kidney Disease: Improving Global Outcomes (KDIGO) Transplant Work Group. KDIGO clinical practice guideline for the care of kidney transplant recipients. Am J Transplant. 2009; 9 Suppl 3:S1-155.

23. Ponticelli C, Locatelli F. Glucocorticoids in the treatment of glomerular diseases: Pitfalls and pearls. *Clin J Am Soc Nephrol.* 2018;13(5):815-22. doi:10.2215/CJN.12991117

24. Dall'Era M, Cisternas MG, Snipes K, Herrinton LJ, Gordon C, Helmick CG. The incidence and prevalence of systemic lupus erythematosus in San Francisco County, California: The California Lupus Surveillance Project. Arthritis Rheumatol. 2017;69(10):1996-2005. doi:10.1002/ art.40191